Author’s response to reviews

Title: The effects of martial arts participation on mental and psychosocial health outcomes: A randomised controlled trial of a secondary school-based mental health promotion program

Authors:

Brian Moore (brian.william.moore@det.nsw.edu.au)
Dean Dudley (dean.dudley@mq.edu.au)
Stuart Woodcock (stuart.woodcock@mq.edu.au)

Version: 1 Date: 28 Oct 2018

Author’s response to reviews:

Dear BMC Psychology,

Thank you for your email.

We have revised the manuscript as requested and resubmitted.

Regarding your queries please see the following:

1. Ethics approval letters

Copies of all relevant ethics approval letters have been forwarded to BMCSeriesEditorial@biomedcentral.com as requested.

2. Funding/peer review

We confirm that the study has not received external funding. The study has undergone peer review and been approved by the hosting institution (i.e. Macquarie University).

3. Study status

We confirm that the study is ongoing. The study is currently collecting data.

Thank you for the opportunity to submit our research to BMC Psychology. Please contact us if you have any queries.

Kind regards
Brian Moore

Macquarie University